Author’s response to reviews

Title: Depression and Anxiety in Patients with Rheumatoid Arthritis: Prevalence rates based on a comparison of the Depression, Anxiety and Stress Scale (DASS) and the Hospital, Anxiety and Depression Scale (HADS)

Authors:

Tanya Covic (t.covic@uws.edu.au)
Steven Cumming (s.cumming@usyd.edu.au)
Julie F Pallant (jpallant@unimelb.edu.au)
Nick Manolios (nick_manolios@wmi.usyd.edu.au)
Paul Emery (P.Emery@leeds.ac.uk)
Philip G Conaghan (P.Conaghan@leeds.ac.uk)
Alan Tennant (A.Tennant@leeds.ac.uk)

Version: 4 Date: 19 January 2012

Author’s response to reviews:

Dear BMC Psychiatry,

Thank you for your instructions with regard the Proof of the paper. I have made slight modifications to the Title, and to the format of the authors names. I have otherwise checked the manuscript, which appears to be fine.

Kind Regards

Alan Tennant